Author's response to reviews

Title: Relationship between Maternal Obesity and Prenatal, Metabolic Syndrome, Obstetrical and Perinatal Complications of Pregnancy in Indiana, 2008-2010

Authors:

Shingairai A Feresu (shinga.feresu@up.ac.za)
Yi Wang (zephyrwang113@gmail.com)
Stephanie Dickinson (sd3@indiana.edu)

Version: 2 Date: 18 January 2015

Author's response to reviews: see over
January 18, 2015

BMC Pregnancy and Childbirth
BioMed Central
236 Gray’s Inn Road
London WC1X 8HB
United Kingdom

I am resubmitting a manuscript titled “Relationship between Maternal Obesity and Prenatal, Metabolic Syndrome, Obstetrical and Perinatal Complications of Pregnancy in Indiana, 2008-2010”, for consideration for publication with your journal. This manuscript is being solely submitted to BMC Pregnancy and Childbirth, and is not under consideration elsewhere until a final decision is made by the editors.

The resubmission contains:

Tables at the end of the manuscript on pages 26-32, space were created between Table heading and the table. The tables are in word and can be edited
An ethics statement has been added at the end of the methods line 96-102.

The study focuses on the relationship between maternal pre-pregnancy obesity and prenatal, metabolic syndrome, obstetrical and perinatal complications for women delivering in Indiana from 2008 through 2010. The study helps understand the obstetric risks of maternal obesity, at a state level, which will be important for improving obstetric care in Indiana and other related and similar states. It also helps to build a body of knowledge which is currently limited in Indiana.

Sincerely,

Prof. Shinga(ira) Feresu, MPH, PhD
Epidemiology & Biostatistics Track
School of Health Systems and Public Health
University of Pretoria
5-10 H.W. Snyman Building
Pretoria, South Africa
Tel: +27 12 354 2376
Fax: +27 12 354 2071
e-mail: shinga.feresu@up.ac.za
Website: http://shsph.up.ac.za